Patient-reported outcome measures for monitoring primary care patients with depression: the PROMDEP cluster RCT and economic evaluation

Tony Kendrick,^{1*} Christopher Dowrick,² Glyn Lewis,³ Michael Moore,¹ Geraldine M Leydon,¹ Adam WA Geraghty,¹ Gareth Griffiths,⁴ Shihua Zhu,¹ Guiqing Lily Yao,⁵ Carl May,⁶ Mark Gabbay,² Rachel Dewar-Haggart,¹ Samantha Williams,¹ Lien Bui,¹ Natalie Thompson,¹ Lauren Bridewell,¹ Emilia Trapasso,² Tasneem Patel,² Molly McCarthy,² Naila Khan,² Helen Page,² Emma Corcoran,³ Jane Sungmin Hahn,³ Molly Bird,³ Mekeda X Logan,³ Brian Chi Fung Ching,³ Riya Tiwari,¹ Anna Hunt¹ and Beth Stuart⁷

- ¹School of Primary Care, Population Health and Medical Education, University of Southampton, Southampton, UK
- ²Department of Primary Care and Mental Health, Institute of Population Health, University of Liverpool, Liverpool, UK
- ³Division of Psychiatry, Faculty of Brain Sciences, University College London, London, UK
- ⁴Southampton Clinical Trials Unit, University of Southampton and University Hospital Southampton NHS Foundation Trust, Southampton, UK
- ⁵Leicester Clinical Trials Unit, University of Leicester, Leicester, UK
- ⁶Faculty of Public Health and Policy, London School of Hygiene and Tropical Medicine, London, UK
- ⁷Centre for Evaluation and Methods, Wolfson Institute of Population Health, Faculty of Medicine and Dentistry, Queen Mary University of London, London, UK

*Corresponding author A.R.Kendrick@southampton.ac.uk

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/PLRQ4216.

Primary conflicts of interest: Tony Kendrick, Christopher Dowrick, Glyn Lewis, Michael Moore, Geraldine Leydon, Adam WA Geraghty, Gareth Griffiths, Shihua Zhu, Guiqing Lily Yao, Carl May, Mark Gabbay and Beth Stuart have received grant funding to their employer universities from the National Institute for Health and Care Research (NIHR) to carry out this study and other research. In addition, Glyn Lewis has received grant funding from the Medical Research Council and Wellcome. Gareth Griffiths has received funding from Janssen-Cilag (High Wycombe, UK), AstraZeneca (Cambridge, UK), Novartis (Basel, Switzerland), Astex (Cambridge, UK), Roche (Basel, Switzerland), HeartFlow (Mountain View, CA, USA), Celldex (Hampton, NJ, USA), BMS, BioNTech (Mainz, Germany), Cancer Research UK (London, UK), the NIHR, the British Lung Foundation (London, UK), Unitaid (Geneva, Switzerland) and GSK (Brentford, UK) for unrelated academic clinical trials and programme funding and has received personal payments from AstraZeneca for delivering Continuing Professional Development training courses. Mark Gabbay has received consultancy fees from Spectrum Learning and Development (Wakefield, UK) as a board member for substance misuse training courses. Tony Kendrick was a member of the NHS England Quality Outcomes Framework Advisory Committee 2009–14, NICE Quality Indicators Advisory Committee 2015–8 and NICE Depression Guideline Update Committee 2015–22 and has been a member of the NHS England Improving Access to Psychological Therapies Expert Advisory Committee since 2020. Christopher Dowrick chaired the WONCA Working Party on Mental Health 2016-21. Guiging Lily Yao is a member of the National Institute for Health and Care Excellence Public Health Committee. Glyn Lewis is a member of the NIHR Efficacy and Mechanism Evaluation Funding Committee. Gareth Griffiths is Director of the Southampton Clinical Trials Unit, which is part-funded by the NIHR. Beth Stuart is a member of the NIHR Health Technology Assessment Commissioning Committee. Tony Kendrick, Glyn Lewis, Michael Moore, Gareth Griffiths, Guiqing Lily Yao and Beth Stuart have been members of trial steering committees for other NIHR-funded studies.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research, which include language which may offend some readers.

Published March 2024 DOI: 10.3310/PLRQ4216

Plain language summary

Patient-reported outcome measures for monitoring primary care patients with depression: the PROMDEP cluster RCT and economic evaluation

Health Technology Assessment 2024; Vol. 28: No. 17 DOI: 10.3310/PLRQ4216

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Depression is common, can be disabling and costs the nation billions. The National Health Service Precommends general practitioners who treat people with depression use symptom questionnaires to help assess whether those people are getting better over time. A symptom questionnaire is one type of patient-reported outcome measure. Patient-reported outcome measures appear to benefit people having therapy and mental health care, but this approach has not been tested thoroughly in general practice. Most people with depression are treated in general practice, so it is important to test patientreported outcome measures there, too.

In this study, we tested whether using a patient-reported outcome measure helps people with depression get better more quickly. The study was a 'randomised controlled trial' in general practices, split into two groups. In one group, people with depression completed the Patient Health Questionnaire, or 'PHQ-9', patient-reported outcome measure, which measures nine symptoms of depression. In the other group, people with depression were treated as usual without the Patient Health Questionnaire-9. We fed the results of the Patient Health Questionnaire-9 back to the people with depression themselves to show them how severe their depression was and asked them to discuss the results with the practitioners looking after them.

We found no differences between the patient-reported outcome measure group and the control group in their level of depression; their work or social life; their satisfaction with care from their practice; or their use of medicines, therapy or specialist care for depression. However, we did find that their quality of life was improved at 6 months, and the costs of the National Health Service services they used were lower.

Using the Patient Health Questionnaire-9 can improve patients' quality of life, perhaps by making them more aware of improvement in their depression symptoms, and less anxious as a result. Future research should test using a patient-reported outcome measure that includes anxiety and processing the answers through a computer to give practitioners clearer advice on possible changes to treatment for depression.

Health Technology Assessment

ISSN 2046-4924 (Online)

Impact factor: 3.6

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2022 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Manuscripts are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This manuscript

The research reported in this issue of the journal was funded by the HTA programme as award number 17/42/02. The contractual start date was in November 2018. The draft report began editorial review in November 2022 and was accepted for publication in July 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this manuscript.

This manuscript presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2024 Kendrick *et al.* This work was produced by Kendrick *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).